Author's response to reviews

Title: Subclinical iron deficiency is a strong predictor of bacterial vaginosis in early pregnancy

Authors:

Hans Verstraelen (hans.verstraelen@UGent.be)
Joris R Delanghe (joris.delanghe@UGent.be)
Kristien Roelens (kristien.roelens@UGent.be)
Stijn I Blot (stijn.blot@UGent.be)
Geert Claeys (geert.claeys@UGent.be)
Marleen Temmerman (marleen.temmerman@UGent.be)

Version: 3 Date: 4 July 2005

Author's response to reviews: see over
Ghent, July 3 2005

To The Editor
BMC Infectious Diseases

Revised Manuscript Submission

Manuscript nr: MS 1174229054599441

Manuscript title: Subclinical iron deficiency is a strong predictor of bacterial vaginosis in early pregnancy

Corresponding author: Hans Verstraelen, MD, MPH
Department of Obstetrics & Gynaecology, Faculty of Medicine and Health Sciences, Ghent University, Ghent University Hospital 1P3, De Pintelaan 185, B-9000 Ghent, Belgium
+32-9-240 37 96 ☏ +32-9-240 38 31 Fax Hans.Verstraelen@UGent.be
Dear Editor,

Thank you for provisionally considering our manuscript for publication.

Please find attached a second revised version of our manuscript entitled “Subclinical iron deficiency is a strong predictor of bacterial vaginosis in early pregnancy”.

We accounted for all formatting changes requested (see below), as listed in your email dd. June 30, 2005.

We hope that the revised manuscript may now be suitable for publication in the Journal.

Sincerely yours,

(On behalf of all authors)

Hans Verstraelen

(Corresponding author)

----------------------------

Formatting changes requested
----------------------------

**Keywords section** - Please remove the keywords section from the manuscript

[Author] The Keywords section has been removed from the title page in the revised manuscript.

**Equal contributors** - These should be indicated on the title page.

[Author] We decided that there is no equal contributorship that needs to be mentioned.

**References** - References should be cited in the text using consecutive numbers in square brackets.

[Author] References have been numbered consecutively as they appear in the manuscript and the references have been formatted as Arabic numbers in square brackets in the revised manuscript.

**Highlighting/Tracking** - Please remove any highlighting/tracking from your manuscript text.

[Author] Highlighting of Table and Figure legends have been removed (text has been made black) in the revised manuscript.

**Competing interests** - Please amend the ‘Competing interests’ section if there are none to declare, please write ‘The authors declare that they have no competing interests’. 
Page breaks - Please remove page breaks from the manuscript.

Figure 1 - Please remove this from the manuscript file.

Tables - Please place the tables in the manuscript with the table title above each table (there appears to be no tables in the manuscript).

Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure cropping guide may be of some help, and can be viewed here: http://www.biomedcentral.com/info/ifora/cropping.

[Author] This has been changed in the revised manuscript.